Author's response to reviews

Title: Iron repletion is associated with reduction in platelet counts in non-dialysis chronic kidney disease patients independent of erythropoiesis-stimulating agent use: a retrospective cohort study

Authors:

Lenar Yessayan (lyessay1@hfhs.org)
Jerry Yee (jye1@hfhs.org)
Gary Zasuwa (gzasuwa1@hfhs.org)
Stan Frinak (sfrinak1@hfhs.org)
Anatole Besarab (abesarab@Fibrogen.com)

Version: 3
Date: 10 July 2014

Author's response to reviews:

1. Ethical approval

Please provide the full name of the ethical committee that approved your study within this statement in the Methods section.

We have added the following statement in the Methods section
"IRB project number 6650; Henry Ford Health Systems Institutional Review Board"

2. Authors' Contributions

It has come to our attention that there is a potential typographical error in this section. Please clarify who the individual referred to as 'TB' is. If you intend to include an additional author at this late stage we will require written confirmation from all co-authors (plus this individual) agreeing to the change in authorship.

This was a typographical error. TB should be AB. Issue has been corrected.